Author's response to reviews

Title: A mouse model for triple-negative breast cancer tumor-initiating cells (TNBC-TICs) exhibits similar aggressive phenotype to the human disease

Authors:

Punit Kaur (pkaur@medicine.tamhsc.edu)
Ganachari M Nagaraja (Ganachari-Mallappa@medicine.tamhsc.edu)
Hongying Zheng (hzheng@medicine.tamhsc.edu)
Dawit Gizachew (dgizachew@neo.tamu.edu)
Moses Galukande (mosesg@img.co.ug)
Sunil Krishnan (SKrishnan@mdanderson.org)
Alexzander Asea (asea@medicine.tamhsc.edu)

Version: 5 Date: 30 January 2012

Author's response to reviews:

January 30, 2012

Dear Editor BMC Cancer,

The authors thank the Editor for accepting our manuscript entitled “A mouse model for triple-negative breast cancer tumor-initiating cells (TNBC-TICs) exhibits similar aggressive phenotype to the human disease” for publication in the journal BMC Cancer.

The accepted version of the manuscript contains the figure legends (see pages 18-20).

Sincerely,

Alexzander Asea, Ph.D.